Author’s response to reviews

Title: Identification of a rare SEPT9 variant in a family with autosomal dominant Charcot-Marie-Tooth disease

Authors:

Gerrit M. Grosse (grosse.gerrit@mh-hannover.de)

Christine Bauer (Christine-Bauer@gmx.co.uk)

Bruno Kopp (kopp.bruno@mh-hannover.de)

Christoph Schrader (schrader.christoph@mh-hannover.de)

Alma Osmanovic (osmanovic.alma@mh-hannover.de)

Version: 2 Date: 22 Jan 2020

Author’s response to reviews:

Dear colleagues,

we thank the reviewers and the editorial staff for their evaluation of our manuscript. According to the suggestions made, we changed and added the corresponding sections regarding the consents for participating and publishing as well as the statement concerning data availability.

Once we will get all signatures from the authors we will send the appropriate authorship change form to the editorial office.

Yours sincerely,

Gerrit M. Grosse (on behalf of the authors)